Author’s response to reviews

Title: Waist circumference cut-off values for the prediction of cardiovascular risk factors clustering in Chinese school-aged children: a cross-sectional study

Authors:

Ailing Liu (liuailing72@yahoo.com)
Andrew P Hills (a.hills@qut.edu.au)
Xiaoqi Hu (huxiaogee@163.com)
Yanping Li (liyanping72@yahoo.com)
Lin Du (dulinaa@hotmail.com)
Ying Xu (wanghw101@sohu.com)
Nuala M Byrne (n.byrne@qut.edu.au)
Guansheng Ma (mags@chinacdc.net.cn)

Version: 6 Date: 17 February 2010

Author’s response to reviews: see over
Dear Editor,

Thank you very much for your checking my article. According to your request, I checked my manuscript very carefully again and made some minor changes. Detailed below are the changes and I also marked the changes with red word in the manuscript. I would like to confirm here that now the article is ready to publish. I am sorry for any inconvenient.

Best regards

Ailing

**Minor changes in the main text:**

1. In the line 6 of page 8, “factor” has been changed into “factors”.
2. In the line 3 of page 9, “Odds Ratio” had been changed into “odds ratio”.
3. In the line 10 of page 9, “clusting” has been changed into “clustering”.
4. In the 1 reference, a space was deleted before the journal title “Obesity”.